%~MoulinÃ© par MaN_auto v.0.33.5 2025-04-14 16:42:35
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{GÃ©omÃ©trie algÃ©brique, GÃ©omÃ©trie et Topologie}
\TopicEN{Algebraic geometry, Geometry and Topology}

\addbibresource{CRMATH_Zinger_20240265.bib}

\usepackage{mathtools}

\newcommand*\eref[1]{\eqref{#1}}
\newcommand*\C{\mathbb C}
\newcommand*\cC{\mathcal C}
\newcommand*\cM{\mathcal M}
\newcommand*\cO{\mathcal O}
\renewcommand*\P{\mathbb P}
\newcommand*\Q{\mathbb Q}
\newcommand*\R{\mathbb R}
\newcommand*\Z{\mathbb Z}

\newcommand*\fc{\mathfrak c}
\newcommand*\io{\iota}
\newcommand*\om{\omega}
\newcommand*\si{\sigma}
\newcommand*\vr{\varrho}

\newcommand*\Si{\Sigma}

\newcommand*\BL{\mathbb{BL}}

\newcommand*\eset{\emptyset}

\DeclareMathOperator*{\opout}{out}
\DeclareMathOperator*{\opin}{in}
\newcommand*{\opini}{{\textstyle{\opin_{i}}}}

%%%---------------------------------------------------------------------------

%%% ARROWS

%%% \to macro
\renewcommand*{\to}{\mathchoice{\longrightarrow}{\rightarrow}{\rightarrow}{\rightarrow}}
%% Remplacer \xrightarrow{#1} par \overset{#1}{\longrightarrow}. 
%
%%% \mapsto macro
%\let\oldmapsto\mapsto
%\renewcommand*{\mapsto}{\mathchoice{\longmapsto}{\oldmapsto}{\oldmapsto}{\oldmapsto}}
%
%%% \longhookrightarrow macro
%\newcommand{\longhookrightarrow}{\!\lhook\joinrel\longrightarrow}
%
%%% Convergence
%\newcommand\dconverge[2][p \to + \infty]{\xrightarrow[#1]{#2}}
%\newcommand\tconverge[1][p \to + \infty]{\underset{#1}{\longrightarrow}}

%%%---------------------------------------------------------------------------

%%% DELIMITERS

%%\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
%\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
%\DeclarePairedDelimiter{\bracks}{[}{]}
%\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
%\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
%\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

%%% Better widebar macro
\makeatletter
\let\save@mathaccent\mathaccent
\newcommand*\if@single[3]{%
 \setbox0\hbox{${\mathaccent"0362{#1}}^H$}%
 \setbox2\hbox{${\mathaccent"0362{\kern0pt#1}}^H$}%
 \ifdim\ht0=\ht2 #3\else #2\fi
 }
%The bar will be moved to the right by a half of \macc@kerna, which is computed by amsmath:
\newcommand*\rel@kern[1]{\kern#1\dimexpr\macc@kerna}
%If there's a superscript following the bar, then no negative kern may follow the bar;
%an additional {} makes sure that the superscript is high enough in this case:
\newcommand*\widebar[1]{\@ifnextchar^{{\wide@bar{#1}{0}}}{\wide@bar{#1}{1}}}
%Use a separate algorithm for single symbols:
\newcommand*\wide@bar[2]{\if@single{#1}{\wide@bar@{#1}{#2}{1}}{\wide@bar@{#1}{#2}{2}}}
\newcommand*\wide@bar@[3]{%
 \begingroup
 \def\mathaccent##1##2{%
%Enable nesting of accents:
  \let\mathaccent\save@mathaccent
%If there's more than a single symbol, use the first character instead (see below):
  \if#32 \let\macc@nucleus\first@char \fi
%Determine the italic correction:
  \setbox\z@\hbox{$\macc@style{\macc@nucleus}_{}$}%
  \setbox\tw@\hbox{$\macc@style{\macc@nucleus}{}_{}$}%
  \dimen@\wd\tw@
  \advance\dimen@-\wd\z@
%Now \dimen@ is the italic correction of the symbol.
  \divide\dimen@ 3
  \@tempdima\wd\tw@
  \advance\@tempdima-\scriptspace
%Now \@tempdima is the width of the symbol.
  \divide\@tempdima 10
  \advance\dimen@-\@tempdima
%Now \dimen@ = (italic correction / 3) - (Breite / 10)
  \ifdim\dimen@>\z@ \dimen@0pt\fi
%The bar will be shortened in the case \dimen@<0 !
  \rel@kern{0.6}\kern-\dimen@
  \if#31
   \overline{\rel@kern{-0.6}\kern\dimen@\macc@nucleus\rel@kern{0.4}\kern\dimen@}%
   \advance\dimen@0.4\dimexpr\macc@kerna
%Place the combined final kern (-\dimen@) if it is >0 or if a superscript follows:
   \let\final@kern#2%
   \ifdim\dimen@<\z@ \let\final@kern1\fi
   \if\final@kern1 \kern-\dimen@\fi
  \else
   \overline{\rel@kern{-0.6}\kern\dimen@#1}%
  \fi
 }%
 \macc@depth\@ne
 \let\math@bgroup\@empty \let\math@egroup\macc@set@skewchar
 \mathsurround\z@ \frozen@everymath{\mathgroup\macc@group\relax}%
 \macc@set@skewchar\relax
 \let\mathaccentV\macc@nested@a
%The following initialises \macc@kerna and calls \mathaccent:
 \if#31
  \macc@nested@a\relax111{#1}%
 \else
%If the argument consists of more than one symbol, and if the first token is
%a letter, use that letter for the computations:
  \def\gobble@till@marker##1\endmarker{}%
  \futurelet\first@char\gobble@till@marker#1\endmarker
  \ifcat\noexpand\first@char A\else
   \def\first@char{}%
  \fi
  \macc@nested@a\relax111{\first@char}%
 \fi
 \endgroup
}
\makeatother

\let\oldbar\bar
\renewcommand*{\bar}[1]{{\mathchoice{\widebar{#1}}{\widebar{#1}}{\widebar{#1}}{\oldbar{#1}}}}

% /!\ Il y a parfois des soucis avec cette macro lorsqu'il y a des caractÃ¨res en minuscule en subscript (et dans les captions). Dans ce cas, faire avec \overline et des espaces manuels.

%%% Double bar
% \overline{\overline{\!a}}

\newcommand*\ov[1]{\bar{#1}}
\newcommand*{\ovcM}{\bar{\cM}\!\mk}
\newcommand*{\ovcMcapt}{\,\,\,\,\,\overline{\!\!\!\!\!\cM\,\,}\!\!}

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\newcommand*{\relabel}{\renewcommand{\labelenumi}{(\theenumi)}}
\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}\relabel}
\newcommand*{\Romanenumi}{\renewcommand*{\theenumi}{\Roman{enumi}}\relabel}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}\relabel}
\newcommand*{\Alphenumi}{\renewcommand*{\theenumi}{\Alph{enumi}}\relabel}
\let\oldtilde\tilde
\renewcommand*{\tilde}[1]{\mathchoice{\widetilde{#1}}{\widetilde{#1}}{\oldtilde{#1}}{\oldtilde{#1}}}
\let\oldforall\forall
\renewcommand*{\forall}{\mathrel{\oldforall}}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses
% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
 \vglue3mm%
% \hspace*{0pt}%
\begingroup\hsize=13cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
 \leftskip\abstractmarginL
 \normalfont\Small
 \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title{The (co)homology of the Deligne--Mumford moduli spaces of marked rational curves}
\alttitle{La (co)homologie des espaces de moduli de Deligne--Mumford de courbes rationnelles marquÃ©es}

\author{\firstname{Aleksey} \lastname{Zinger}}
\address{Department of Mathematics, Stony Brook University, Stony Brook, NY 11794, United States of America}
\email{azinger@math.stonybrook.edu}

\begin{abstract}
This informal note collects key results and open problems on the (co)homology of the Deligne--Mumford moduli spaces of real marked rational curves. The open problems are both of topological nature, aiming to investigate the (co)homology of these spaces further, and of algebraic nature, aiming to describe the structure of the homology of these spaces in operad-like terms.
\end{abstract}

\begin{altabstract}
Cette note informelle rassemble des rÃ©sultats clÃ©s et des problÃ¨mes ouverts sur la (co)homologie des espaces de moduli de Deligne--Mumford des courbes rationnelles marquÃ©es rÃ©elles. Les problÃ¨mes ouverts sont Ã  la fois de nature topologique, visant Ã  approfondir l'Ã©tude de la (co)homologie de ces espaces, et de nature algÃ©brique, visant Ã  dÃ©crire la structure de l'homologie de ces espaces en termes d'opÃ©rades.
\end{altabstract}

\begin{document}

\maketitle

This note includes a readily accessible presentation of the recent results of~\cite{RDMhomol,RDMbl} on the topology of real Deligne--Mumford moduli spaces. These results are the main outcome of a decade-old aim, the realization of which also includes the systematic treatment of blowups in~\cite{RDMbl2}, and are the main topological inputs in~\cite{RGWsAlg}. However, the primary motivation for this note is an algebraic reformulation of the results of~\cite{RDMhomol,RDMbl}, in terms of operads, which was inspired by~\cite{EHKR} and the analogous reformulation of the main results of~\cite{Keel} in~\cite{Getz95}. We also include open problems, of both purely topological nature and purely algebraic, with the latter based on the operad formalism.

Below are the two moduli spaces central to this note.

$\ovcM_{\ell}=$ Deligne--Mumford moduli space of (complex) rational curves with $\ell$~marked points; smooth projective variety of complex dimension~$\ell-3$. An element of this space is the equivalence class of stable tuples
\[
\cC\equiv\parens{\Si,z_1,\dots,z_{\ell}},
\]
where $\Si$ is a connected, possibly nodal, Riemann surface and $z_1,\dots,z_{\ell}\in \Si$ are distinct smooth points.

$\R\ovcM_{k,\ell}=$ Deligne--Mumford moduli space of real rational curves with $k$~real marked points and $\ell$ conjugate pairs of marked points; compact smooth real analytic variety of dimension~$k+2\ell-3$; orientable if $k = 0$ or $k+2\ell \le 4$. An element of this space is the equivalence class of stable tuples
\[
\cC\equiv\parens[\big]{\Si,\si,z_1,\dots,z_k,(z_1^+,z_1^-),\dots,(z_{\ell}^+,z_{\ell}^-)},
\]
where $\Si$ is a connected, possibly nodal, Riemann surface, $\si$ is an anti-holomorphic involution on~$\Si$ and
\[
z_1,\dots,z_k,z_1^+,z_1^-,\dots,z_{\ell}^+,z_{\ell}^-\in\Si
\]
are distinct smooth points such that $\si(z_i^+) = z_i^-$ for $i = 1,\dots,\ell$.

For $\ell\in \Z^{\ge0}$, we define $[\ell] = \{1,\dots,\ell\}$.

\section{Key results}

\begin{proof}[Complex case, $\ovcM_{\ell}$]
This smooth projective variety contains smooth divisors~$D_{J,K}$, with $J \sqcup K = [\ell]$ and $\abs{J},\abs{K} \ge 2$, whose generic element is a two-component curve with the marked points on the two components indexed by~$J$ and~$K$; see Figure~\ref{M04rel_fig} for some examples. It is a now classical result of Keel~\cite{Keel} that
\begin{enumerate}\renewcommand{\theenumi}{\ensuremath{\C}\arabic{enumi}}\relabel
\item\label{Cgen_it} these divisors generate $H^*(\ovcM_{\ell};\Z)$ as an algebra,
\item\label{Crel_it} subject only to some obvious relations;
\end{enumerate}
see~\cite[Theorem~2.1]{RDMhomol} for a precise statement. The relations arise from
\begin{enumerate}\renewcommand{\theenumi}{\ref*{Crel_it}\alph{enumi}}\relabel
\item certain divisors~$D_{J,K} \subset \ovcM_{\ell}$ being disjoint and
\item all divisors~$D_{J,K}$ in~$\ovcM_4 \approx \C\P^1$ being points and thus equivalent in $H^2(\ovcM_4;\Z)$, as indicated in Figure~\ref{M04rel_fig}.
\end{enumerate}
The above relations in $H^2(\ovcM_4;\Z)$ pull back to relations in $H^2(\ovcM_{\ell};\Z)$ with $\ell \ge 4$ via the forgetful morphisms $\ovcM_{\ell} \to \ovcM_4$ dropping all but four of the marked points (every 4-element subset of~$[\ell]$ determines such a morphism).

\begin{figure}[h]
\includegraphics[width=.6\linewidth]{M04rel_fig}
\caption{Relations between the divisors $D_{J,K} \subset \ovcMcapt_4$ in $H^2(\ovcMcapt_4;\Z)$.\label{M04rel_fig}}
\end{figure}

The proof in~\cite{Keel} presents $\ovcM_{\ell+1}$ as a sequence of holomorphic blowups of~$\ovcM_{\ell} \times \ovcM_4$ and determines $H^*(X_{\vr};\Z)$ inductively for every intermediate blowup~$X_{\vr}$, assuming the claimed result for~$H^*(\ovcM_{\ell};\Z)$. The proof of the main result of~\cite{Keel} given in~\cite{RDMhomol} still uses this sequence of blowups to obtain~\eqref{Cgen_it}, but establishes~\eqref{Crel_it} directly, without determining $H^*(X_{\vr};\Z)$ for the intermediate blowups~$X_{\vr}$.

For any commutative ring $R$ with unity~1, the modules $H_*(\ovcM_{\ell};R)$ can be arranged into an operad of graded $R$-modules
\[
\cO_{\C}(\ell)\equiv
\begin{cases}
R,						& \text{if} \ \ell = 1;\\
H_*(\ovcM_{\ell+1};R),	& \text{if} \ \ell \ge 2;
\end{cases}
\]
as follows. For integers $k,\ell \ge 2$ and $i\in [k]$, let
\[
\io_{k\ell i} \colon \ovcM_{k+1} \times \ovcM_{\ell+1} \to \ovcM_{k+\ell}
\]
be the immersion identifying the $i$-th marked point of a marked curve $[\cC_1]\in \ovcM_{k+1}$ with the last marked point of a marked curve $[\cC_2]\in \ovcM_{\ell+1}$, keeping the indices of the first $i-1$ marked points of~$[\cC_1]$ the same, increasing the indices of the last $k+1-i$ marked points of~$[\cC_1]$ by~$\ell-1$, and increasing the indices of the first $\ell$~marked points of~$[\cC_2]$ by~$i-1$. This induces homomorphisms
\[
\io_{k\ell i*} \colon \cO_{\C}(k) \times \cO_{\C}(\ell) \approx H_*\parens[\big]{\ovcM_{k+1} \times \ovcM_{\ell+1}; R} \to \cO_{\C}(k + \ell - 1).
\]
For $k,\ell\in \Z^+$ and $i\in [k]$, define
\begin{equation}\label{prodcases_e}
\begin{aligned}
\io_{k1i*} \colon \cO_{\C}(k) \times \cO_{\C}(1) & \to \cO_{\C}(k), 			\quad 	&& \io_{k1i*}(\eta,r) = r\eta,\\
\io_{1\ell 1*} \colon \cO_{\C}(1) \times \cO_{\C}(\ell) & \to \cO_{\C}(\ell),  		&& \io_{1\ell 1*}(r,\eta) = r\eta.
\end{aligned}
\end{equation}
For $k,\ell_1,\dots,\ell_k\in \Z^+$, define
\begin{equation}\label{expandcomp_e}
\begin{gathered}
\circ \colon \cO_{\C}(k) \times \cO_{\C}(\ell_1) \times \dots \times \cO_{\C}(\ell_k) \to \cO_{\C}\parens{\ell_1 + \dots + \ell_k},\\
\eta \circ (\eta_1,\dots,\eta_k) =\io_{(1+\ell_2+\dots+\ell_k)\ell_11*}
\parens[\bigg]{\cdots\io_{(k-1+\ell_k)\ell_{k-1}(k-1)*}\parens[\big]{\io_{k\ell_k k*}(\eta,\eta_k),\eta_{k-1}},
\dots,\eta_1}.
\end{gathered}
\end{equation}
The resulting operad $\parens[\big]{\parens{\cO_{\C}(\ell)}_{\ell\in\Z^+},\circ}$ is the operad of hypercommutative algebras; see~\cite{Getz95}.
\let\qed\relax
\end{proof}

\begin{proof}[Real case with real points only, $\R\ovcM_{k,0}$]
By the main result of~\cite{Krasnov}, $H^*(\R\ovcM_{k,0};\Z_2)$ is described in the same way as $H^{2*}(\ovcM_k;\Z)$ above with the complex subvarieties $D_{J,K} \subset \ovcM_k$ replaced by the analogous real subvarieties
$\R D_{J,K} \subset \R\ovcM_{0,k}$; see the proof of~\cite[Theorem~5.6]{EHKR}. Since $\R\ovcM_{k,0}$ is not orientable for $k \ge 5$, the homology and cohomology of~$\R\ovcM_{k,0}$ with coefficients in a ring of characteristic larger than~2 are quite distinct. For any commutative ring~$R$ with unity and distinct $a,b,c,d\in [k]$, $H^1(\R\ovcM_{k,0};R)$ contains the pullback~$\om_{abcd}$ of the PoincarÃ© dual of a point in $\R\ovcM_{4,0} \approx \R\P^1$ by the forgetful morphism
$\R\ovcM_{k,0} \to \R\ovcM_{4,0}$ sending the marked points indexed by $a$, $b$, $c$, $d$ to the marked points indexed by~$1$, $2$, $3$, $4$, respectively, and dropping the remaining marked points. This class is PoincarÃ© dual to a linear combination of the real hypersurfaces~$\R D_{J,K}$.

By~\cite[Theorem~2.9]{EHKR},
\begin{enumerate}\renewcommand{\theenumi}{\ensuremath{\R\R}\arabic{enumi}}\relabel
\item\label{RRgen_it} the classes~$\om_{abcd}$ generate $H^*(\R\ovcM_{k,0};\Q)$ as an algebra,
\item\label{RRrel_it} subject only to some expected relations.
\end{enumerate}
The relations arise from
\begin{enumerate}\renewcommand{\theenumi}{\ref*{RRrel_it}\alph{enumi}}\relabel
\item the diffeomorphism of $\R\ovcM_{4,0}$ induced by the interchange of two marked points being orientation-reversing and
\item $\R\ovcM_{5,0}$ being diffeomorphic to the blowup of $\R\P^1 \times \R\P^1$ at three points of the diagonal.
\end{enumerate}
The resulting relations in $H^1(\R\ovcM_{4,0};\Q)$, $H^1(\R\ovcM_{5,0};\Q)$, and $H^2(\R\ovcM_{5,0};\Q)$ pull back to relations in $H^1(\R\ovcM_{k,0};\Q)$ and $H^2(\R\ovcM_{k,0};\Q)$ via the forgetful morphisms
\[
\R\ovcM_{k,0} \to \R\ovcM_{4,0} \qquad\text{and}\qquad \R\ovcM_{k,0} \to \R\ovcM_{5,0}
\]
dropping all but four (resp.~five) of the marked points. By~\cite[Corollary~6.9]{EHKR}, certain natural subvarieties of~$\R\ovcM_{k,0}$ form a linear basis for~$H_*(\R\ovcM_{k,0};\Q)$. By~\cite[Corollary~3.8]{Rains}, $H_*(\R\ovcM_{k,0};\Z)$ and $H^*(\R\ovcM_{k,0};\Z)$ have only 2-torsion; it is described in~\cite[Section~5]{EHKR}.

Similarly to the complex case, there are node-identifying immersions
\[
\io_{k\ell i}^{\R} \colon \R\ovcM_{k+1,0} \times \R\ovcM_{\ell+1,0} \to \R\ovcM_{k+\ell,0}.
\]
They again determine an operad structure on the sequence
\[
\cO_{\R\R}(k)\equiv
\begin{cases}
\Q,							& \text{if} \ k = 1;\\
H_*(\R\ovcM_{k+1,0};\Q),		& \text{if} \ k \ge 2;
\end{cases}
\]
of graded vector spaces. By~\cite[Theorem~2.14]{EHKR}, this operad is the operad of unital 2-Gerstenhaber algebras.

The proof of~\cite[Corollary~3.8]{Rains} is based on presenting~$\R\ovcM_{k,0}$ as a sequence of real blowups of~$(\R\P^1)^{k-3}$ and shows that $H_*(X_{\vr};\Z)$ has only 2-torsion for every intermediate blowup~$X_{\vr}$. Theorems~2.9 and~2.12 and Corollary~6.9 in~\cite{EHKR} are proved in parallel, by combining the aforementioned topological results of~\cite{Keel,Krasnov} with algebraic considerations. The same results should be obtainable by presenting~$\R\ovcM_{k+1,0}$ as a sequence of real blowups of~$\R\ovcM_{k,0} \times \R\ovcM_{4,0}$ as in~\cite{RDMbl} and relating the (co)homologies of the intermediate blowups as in~\cite[Section~3.2]{RDMhomol}.
\let\qed\relax
\end{proof}

\begin{proof}[Real case with conjugate points only, $\R\ovcM_{0,\ell}$]
This orientable compact smooth manifold contains compact real hypersurfaces~$\R E_{J,K}$, with $J \sqcup K = [\ell]$, and~$\R H_{J,K}$, with $J \sqcup K = [\ell]$ and $J,K \neq \eset$, whose generic elements are curves with two irreducible components that are interchanged by the involution in the first case and are preserved by~it in the second case. This manifold also contains compact submanifolds~$\R D_{I;J,K}$, with $I \sqcup J \sqcup K = [\ell]$, $I \neq \eset$, and $\abs{J}+\abs{K} \ge 2$, of (real) codimension~2 whose generic elements are curves with three components two of which are interchanged by the involution; see Figure~\ref{TopolRel_fig} for some examples. The submanifolds~$\R E_{J,K}$ and~$\R D_{I;J,K}$ are orientable, while~$\R H_{J,K}$ is not if $k \ge 3$.

\begin{figure}[h]
\includegraphics[width=.9\linewidth]{TopolRel_fig}
\caption{The first line represents an equivalence of two points in $H^1(\R\ovcMcapt_{0,2};\Q)$ and a relation between three loops in $H^2(\R\ovcMcapt_{0,3};\Q)$. The bottom diagram represents the intersection pattern of the six codimension~2 submanifolds $\R D_{I;J,K} \approx S^1$ with the four real hypersurfaces $\R E_{J',K'} \approx S^2$ in~$\R\overline\cM_{0,3}$. The former are labeled by the unique element of $I \subset [3]$ and the sign of $(-1)^{\abs{J}} = (-1)^{\abs{K}}$; the latter are labeled by the subset $J',K' \subset [3]$ not containing~1.\label{TopolRel_fig}}
\end{figure}

By~\cite[Theorem~2.2]{RDMhomol},
\begin{enumerate}\renewcommand{\theenumi}{\ensuremath{\R\C}\arabic{enumi}}\relabel
\item\label{RCgen_it} the submanifolds~$\R E_{J,K}$ and~$\R D_{I;J,K}$ generate $H^*(\R\ovcM_{0,\ell};\Q)$ as an algebra,
\item\label{RCrel_it} subject only to some expected relations.
\end{enumerate}
The relations arise from
\begin{enumerate}\renewcommand{\theenumi}{\ref*{RCrel_it}\alph{enumi}}\relabel
\item certain submanifolds $\R E_{J,K},\R D_{I;J,K} \subset \R\ovcM_{0,\ell}$ being disjoint,
\item all real hypersurfaces~$\R E_{J,K}$ in~$\R\ovcM_{0,2} \approx \R\P^1$ being points and thus equivalent in $H^1(\R\ovcM_{0,2};\Q)$, as indicated by the left identity in Figure~\ref{TopolRel_fig}, and
\item the relation in $H_1(\R\ovcM_{0,3};\Q)$, or equivalently in $H^2(\R\ovcM_{0,3};\Q)$, of Remark~3.4 in~\cite{RealEnum}, as indicated by the right identity and the bottom diagram in Figure~\ref{TopolRel_fig}.
\end{enumerate}
The above relations in $H^1(\R\ovcM_{0,2};\Q)$ and $H^2(\R\ovcM_{0,3};\Q)$ pull back to relations in $H^1(\R\ovcM_{0,\ell};\Q)$ and $H^2(\R\ovcM_{0,3};\Q)$ via the forgetful morphisms
\[
\R\ovcM_{0,\ell} \to \R\ovcM_{0,2} \qquad\text{and}\qquad \R\ovcM_{0,\ell} \to \R\ovcM_{0,3}
\]
dropping all but two (resp.~three) conjugate pairs of marked points.

The proof of~\eqref{RCgen_it} in~\cite{RDMhomol} is based on the presentation of $\R\ovcM_{0,\ell+1}$ in~\cite{RDMbl} as a sequence of blowups of~$\R\ovcM_{0,\ell} \times \ovcM_4$ of three different types and describes generators for~$H^*(X_{\vr};\Q)$ for every intermediate blowup~$X_{\vr}$, assuming~\eqref{RCgen_it} for~$\R\ovcM_{0,\ell}$. This approach can also be used to describe generators for~$H_*(\R\ovcM_{0,\ell};\Z)$ and implies that this module has only 2-torsion. The proof of~\eqref{RCrel_it} in~\cite{RDMhomol} relates $H^*(\R\ovcM_{0,\ell+1};\Q)$ to $H^*(\R\ovcM_{0,\ell'};\Q)$ with $\ell' \le \ell$ and $H^*(\ovcM_{\ell'};\Q)$ with $\ell' \le \ell+1$ by pairing certain cohomology classes with the submanifolds~$\R E_{J,K}$ and~$\R D_{I;J,K}$ to show the algebraic independence of the former.

Let $R$ be a commutative ring with unity~1 and
\[
\cO_{\R\C}(k)=H_*\parens{\R\ovcM_{0,k+1};R} \quad \forall k\in \Z^+.
\]
For integers $k \ge 1$, $\ell \ge 2$, and $i\in [k]$, let
\[
\io_{k\ell i}^+ \colon \R\ovcM_{0,k+1} \times \ovcM_{\ell+1} \to \R\ovcM_{0,k+\ell}
\]
be the immersion identifying the first marked point in the $i$-th conjugate pair of a marked curve $[\cC_1]\in \R\ovcM_{0,k+1}$ with the last marked point of a marked curve $[\cC_2]\in \ovcM_{\ell+1}$, and the second marked point in this pair with the last marked point of the conjugate marked curve~$[\ov{\cC_2}]$, keeping the indices of the first $i-1$ conjugate pairs of marked points of~$[\cC_1]$ the same, increasing the indices of the last $k+1-i$ conjugate pairs of marked points of~$[\cC_1]$ by~$\ell-1$, and turning the first $\ell$~marked points of~$[\cC_2]$ into the first marked points of the conjugate pairs with the indices increased by~$i-1$; see Figure~\ref{ioplus_fig} for an example. This induces homomorphisms
\[
\io_{k\ell i*}^+ \colon \cO_{\R\C}(k) \times \cO_{\C}(\ell) \approx H_*\parens[\big]{\R\ovcM_{0,k+1} \times \ovcM_{\ell+1};R} \to \cO_{\R\C}(k+\ell-1).
\]
For $k\in \Z^+$ and $i\in [k]$, define
\[
\io_{k1i*}^+ \colon \cO_{\R\C}(k) \times \cO_{\C}(1) \to \cO_{\R\C}(k),
\qquad \io_{k1i*}^+(\eta,r)=r\eta.
\]
For $k,\ell_1,\dots,\ell_k\in \Z^+$, define
\[
\begin{gathered}
\circ \colon \cO_{\R\C}(k) \times \cO_{\C}(\ell_1) \times \dots \times \cO_{\C}(\ell_k) \to \cO_{\R\C}(\ell_1 + \dots + \ell_k),\\
\eta \circ (\eta_1,\dots,\eta_k) =\io_{(1+\ell_2+\dots+\ell_k)\ell_11*}^+ \parens[\bigg]{\cdots\io_{(k-1+\ell_k)\ell_{k-1}(k-1)*}^+\parens[\big]{\io_{k\ell_kk*}^+(\eta,\eta_k),\eta_{k-1}},\dots,\eta_1}.
\end{gathered}
\]
This determines a right module structure on $\parens{\cO_{\R\C}(k)}_{k\in\Z^+}$ over the operad $\parens[\big]{(\cO_{\C}(\ell)}_{\ell\in\Z^+},\circ)$.
\let\qed\relax
\end{proof}

\begin{figure}[h]
\includegraphics[width=.7\linewidth]{ioplus_fig}
\caption{The action of the immersion~$\io_{232}^+$ on a typical element in its domain; each double-headed arrow labeled~$\si$ indicates the involution on the corresponding real curve.\label{ioplus_fig}}
\end{figure}

\section{Open problems}

\begin{proof}[Real case with conjugate points only, $\R\ovcM_{0,\ell}$]
The approach of~\cite{RDMbl,RDMhomol} to~\eqref{RCgen_it} implies that $H^*(\R\ovcM_{0,\ell};\Z)$ is linearly generated by orientable intersections of the submanifolds $\R E_{J,K}$, $\R H_{J,K}$, and~$\R D_{I;J,K}$ and contains only 2-torsion. For example, the torsion of $H_1(\R\ovcM_{0,3};\Z)$ is generated by $\R H_{\{1,2\},\{3\}} \cap \R H_{\{1\},\{2,3\}}$, as shown in~\cite[Section~3]{RealEnum}. However, it still remains to determine the amount of 2-torsion and the ring structure of~$H^*(\R\ovcM_{0,\ell};\Z)$ for $\ell \ge 4$. The algebraic generators for the torsion of~$H_*(\R\ovcM_{0,\ell};\Z)$ should correspond to trees, with the univalent and bivalent vertices labeled by nonempty subsets of~$[\ell]$ forming a partition of~$[\ell]$. A more algebraic question is to describe the structure of the right module structure~$\parens{\cO_{\R\C}(k)}_{k\in\Z^+}$ over the operad $\parens[\big]{(\cO_{\C}(\ell))_{\ell\in\Z^+},\circ}$ explicitly, in the spirit of~\cite[Theorem~2.14]{EHKR} and~\cite{Getz95}.
\let\qed\relax
\end{proof}

\begin{proof}[The general real case, $\R\ovcM_{k,\ell}$ with $k,\ell\in \Z^+$]
The approach of~\cite{RDMbl,RDMhomol} to~\eqref{RCgen_it} also implies that $H_*(\R\ovcM_{k,\ell};\Z)$ with $k \ge 1$ is linearly generated by orientable intersections of the submanifolds $\R H_{J,K}$ and~$\R D_{I;J,K}$ and contains only 2-torsion. The former are now indexed by pairs $J \equiv (J_{\R},J_{\C})$ and $K \equiv (K_{\R},K_{\C})$~with
\[
[k] = J_{\R} \sqcup K_{\R},
\qquad [\ell] = J_{\C} \sqcup K_{\C},
\qquad \text{and}
\qquad \abs{J_{\R}} + 2\abs{J_{\C}}, \ \abs{K_{\R}} + 2\abs{K_{\C}}\ge 2,
\]
and the latter by the partitions $[\ell] = I \sqcup J \sqcup K$ with $\abs{J}+\abs{K} \ge 2$. Similarly, $H^*(\R\ovcM_{k,\ell};\Z)$ is generated by some linear combinations of these intersections. In fact, the most delicate type of blowups appearing in~\cite{RDMbl,RDMhomol} no longer enters into consideration. However, it remains to determine the amount of 2-torsion in~$H_*(\R\ovcM_{k,\ell};\Z)$ and the ring structure even of~$H^*(\R\ovcM_{k,\ell};\Q)$.

For $k\in \Z^+$ and $\ell\in \Z^{\ge0}$, let
\[
\cO_{\R}(k,\ell) =
\begin{cases}
\Q,								& \text{if} \ k+\ell = 1;\\
H_*(\R\ovcM_{k+1,\ell};\Q),		& \text{if} \ k+\ell \ge 2;
\end{cases}
\qquad\qquad \cO_{\R}(k)=\bigoplus_{\ell=0}^{\infty} \cO_{\R}(k,\ell).
\]
Similarly to the previous cases, there are node-identifying immersions
\[
\io_{k\ell k'\ell' i}^{\R} \colon \R\ovcM_{k+1,\ell} \times \R\ovcM_{k'+1,\ell'} \to \R\ovcM_{k+k',\ell+\ell'}
\]
and
\[
\io_{k\ell\ell' i}^+ \colon \R\ovcM_{k+1,\ell} \times \ovcM_{\ell'+1} \to \R\ovcM_{k+1,\ell+\ell'-1},
\]
with $k+\ell \ge 2$ in both cases, $i\in [k+1]$ and $k'+\ell' \ge 2$ in the first case, and $i\in [\ell]$ and $\ell' \ge 2$ in the last case. They induce homomorphisms
\[
\io_{k\ell k'\ell' i*}^{\R} \colon \cO_{\R}(k,\ell) \times \cO_{\R}(k',\ell') \approx  H_*\parens[\big]{\R\ovcM_{k+1,\ell} \times \R\ovcM_{k'+1,\ell'};\Q} \to \cO_{\R}(k+k'-1,\ell+\ell')
\]
and
\[
\io_{k\ell\ell' i*}^+ \colon \cO_{\R}(k,\ell) \times \cO_{\C}(\ell') \approx  H_*\parens[\big]{\R\ovcM_{k+1,\ell} \times \ovcM_{\ell'+1};\Q} \to \cO_{\R}(k,\ell+\ell'-1).
\]
We define the homomorphisms
\begin{align*}
& \io_{k\ell10i*}^{\R} \colon \cO_{\R}(k,\ell) \times \cO_{\R}(1,0) \to \cO_{\C}(k,\ell),\\
& \io_{10k'\ell' 1*}^{\R} \colon \cO_{\R}(1,0) \times \cO_{\R}(k',\ell') \to \cO_{\C}(k',\ell'),\\
\text{and}\quad
& \io_{k\ell 1 i*}^+ \colon \cO_{\R}(k,\ell) \times \cO_{\C}(1) \to \cO_{\C}(k,\ell)
\end{align*}
similarly to~\eref{prodcases_e}. These bilinear maps induce multilinear maps
\[
\circ \colon \cO_{\R}(k,\ell) \times \cO_{\R}(m_1,\ell_1') \times \dots \times \cO_{\R}(m_k,\ell_k') \to \cO_{\R}(m_1 + \dots + m_k,\ell + \ell_1' + \dots + \ell_k')
\]
and
\[
\circ \colon \cO_{\R}(k,\ell) \times \cO_{\C}(m_1) \times \dots \times \cO_{\C}(m_{\ell}) \to 
\cO_{\R}(k,m_1 + \dots + m_{\ell})
\]
similarly to~\eref{expandcomp_e}. The last two maps determine an operad $\parens{\cO_{\R}(k)}_{k\in\Z^+}$ of graded vector spaces and a right module structure on $\parens{\cO_{\R}(k,\ell)}_{\ell\in\Z^+}$ over the operad $\parens[\big]{\parens{\cO_{\C}(\ell)}_{\ell\in\Z^+},\circ}$ for each $k\in \Z^+$.

As suggested by A.~Voronov, the above structures can be combined into an object resembling a bicolored operad; see~\cite[Section~1.1]{Gir}. For $n\in \Z^+$, let
\[
\cO_{\R}'(n) = \cO_{\C}(n) \sqcup \bigsqcup_{\substack{k+\ell=n \\ k\ge0, \ \ell\ge1}} \cO_{\R}(k+1,\ell-1).
\]
For each fixed $n\in \Z^+$, auxiliary grading is provided by $k\in \Z^{\ge0}$ above, with the elements of~$\cO_{\C}(n)$ being of the auxiliary degree~0. For $x\in \cO_{\R}'(n)$, let $\abs{x} = n$ and
\[
\abs{x}_+=
\begin{cases}
n,			& \text{if} \ x\in \cO_{\C}(n);\\
\ell-1,		& \text{if} \ x\in \cO_{\R}(k+1,\ell-1).
\end{cases}
\]
We take the set of colors to be $\cC \equiv \{+,\R\}$ and define
\[
\opout \colon \cO_{\R}'(n) \to \cC,
\qquad \opout(x) =
\begin{cases}
+,		& \text{if} \ x\in \cO_{\C}(n);\\
\R,		& \text{if} \ x\in \cO_{\R}(k+1,\ell-1).
\end{cases}
\]
For $i\in [n]$, define
\[
\opini \colon \cO_{\R}'(n) \to \cC,
\qquad \opini(x) =
\begin{cases}
+,		& \text{if} \ i \le \abs{x}_+;\\
\R,		& \text{if} \ i > \abs{x}_+.
\end{cases}
\]
Let $1_+ = 1\in \cO_{\C}(1)$ and $1_{\R} = 1\in \cO_{\R}(1,0)$. The above auxiliary grading is preserved by the~maps
\[
\circ_i \colon \braces[\big]{(x,y)\in \cO_{\R}'(m) \times \cO_{\R}'(n) \;:\; \opini(x) = \opout(y)} \to \cO_{\R}'(m+n-i),
\qquad i\in [m],
\]
induced by the homomorphisms~$\io_{k\ell k'\ell' i*}^{\R}$ and $\io_{k\ell\ell' i*}^+$ defined in the previous paragraph. These maps satisfy a modification of~\cite[(1.1.4a)]{Gir},
\begin{equation}\label{114a_e}
x \circ_i (y \circ_j z) =
\begin{cases}
(x \circ_i y) \circ_{j+i-1} z,			& \text{if} \ \opout(y) = \opout(z);\\
(x \circ_i y) \circ_{j+\abs{x}_+} z,		& \text{if} \ \opout(y) \neq \opout(z);
\end{cases}
\qquad\qquad \text{if} \ 1\le i\le \abs{x},\ 1\le j\le \abs{y}.
\end{equation}
These maps also satisfy a modification of~\cite[(1.1.4b)]{Gir},
\begin{equation}\label{114b_e}
(x \circ_i y) \circ_{j+\abs{y}-1} z =
\begin{cases}
(x \circ_j z) \circ_i y,				& \text{if} \ \opout(y) = +;\\
(x \circ_j z) \circ_{i+\abs{z}_+} y,	& \text{if} \ \opout(y) = \R;
\end{cases}
\qquad\qquad \text{if} \ 1\le i< j\le \abs{x}.
\end{equation}
The identities~\eref{114a_e} and~\eref{114b_e} hold whenever both sides are defined. They agree with~\cite[(1.1.4a)]{Gir} and~\cite[(1.1.4b)]{Gir} whenever~$\opout(y) = \opout(x)$ and $\opout(y) = \R$, respectively. The identity~\cite[(1.1.4c)]{Gir} holds without any modifications.
\let\qed\relax
\end{proof}

A natural question is to describe the above algebraic structures explicitly and perhaps to find a more succinct way of presenting them.

\section*{Acknowledgments}

The author would like to thank A.~Voronov for discussions on operads.

\section*{Declaration of interests}

The author does not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and has declared no affiliations other than their research organizations.

\printbibliography
\end{document}